Author's response to reviews

Title: Predictors of mortality among elderly people living in a south Indian urban community; a 10/66 Dementia Research Group prospective population-based cohort study

Authors:

A T Jotheeswaran (Jotheeswaran.amuthavalli_thiyagarajan@kcl.ac.uk)
Joseph D Williams (idw@vsnl.com)
Martin J Prince (martin.prince@kcl.ac.uk)

Version: 7 Date: 22 June 2010

Author's response to reviews:

Two minor edits only (typographical errors)
Alzheimer Disease International = Alzheimer's Disease International in the list of abbreviations
Physically activity = physical activity in one of the tables
Both errors now rectified in resubmitted version
Otherwise everything looks OK